(1) In case history, kindly elaborate on the chief complaint and history of present illness. Please mention other associated symptoms.

**Answer:** Details of the chief complaints and history of present illness have been supplemented in the article. Relevant additional symptoms have been provided.

(2) Modify personal and medical history. There is only a mention of no history of carcinosarcoma. Carcinosarcoma of which site is the author referring to? And why was this mentioned in the history when the diagnosis of the present case is not yet made.

**Answer:** Personal and family histories have been revised. No one in the family had a similar disease. At this time, a definitive diagnosis of the patient's illness has not been established.

(3) Typographical errors in the Physical examination to be rectified.

**Answer:** A typographical error in the physical examination was corrected.

(4) Please mention the provisional and differential diagnosis.

**Answer:** Provisional and differential diagnoses have been supplemented in the article.

(5) Please elaborate on the FNAC findings to indicate on what basis was osteosarcoma considered. Knowing that parotid gland is an unusual site for osteosarcoma.

**Answer:** The results of FNAC and immunohistochemical staining have been described in detail in the article. Immunohistochemical results showed that the tumor cells were positive for SATB2, MDM2, CDK4 and vimentin.

(6) Why as an incisional biopsy not performed?

**Answer:** Cytologic aspiration was performed at another hospital; incisional biopsy was not performed. After admission to our hospital, radical surgery was performed after the relevant examinations were completed.

(7) Was neck dissection and removal of cervical lymph nodes a part of the surgical intervention? In the pathological description of the excised specimen, it is mentioned as "No metastatic tumor was found in cervical lymph node". Please clarify.
Neck dissection and neck lymph-node dissection were not performed. Enlarged lymph nodes were found around the mass in the excised specimen, and no metastatic tumor was found in the lymph nodes. It has been revised in the article.

(8) In fig 3, the legend mentions as Carcinosarcomas gross specimen. This is again incorrect. It can me rephrased as Gross/Macroscopic findings of the excised specimen.

In the article, the legend of Figure 3 has been rephrased as Gross findings of the excised specimen.